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Detector Issues 
 Finally! A quiet week! 

 Couple DAQ resets 

 Tasks performed during beam 

downtime 

 Warm channel swap out at ND 

 CI special run performed at FD 
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MINOS+ Status 
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Near Detector Far Detector 

Start Date/Time End Date/Time POT Fraction Live Time Fraction POT Fraction Live Time Fraction 

12/28/15 00:00:00 1/4/16 00:00:00 81.0% 85.0% 70.0% 71.0% 

1/4/16 00:00:00 1/11/16 00:00:00 97.5% 97.1% 99.8% 99.8% 

1/11/16 00:00:00 1/18/16 00:00:00 81.0% 71.5% 92.1% 96.5% 

1/18/16 00:00:00 1/25/16 00:00:00 97.1% 97.3% 99.7% 99.8% 

1/25/16 00:00:00 2/1/16 00:00:00 98.8% 97.2% 98.7% 96.5% 

2/1/16 00:00:00 2/8/16 00:00:00 99.1% 98.8% 87.2% 86.0% 

2/8/16 00:00:00 2/15/16 00:00:00 99.4% 88.2% 99.2% 88.3% 

2/15/16 00:00:00 2/22/16 00:00:00 85.6% 86.8% 95.4% 96.1% 

2/22/16 00:00:00 2/29/16 00:00:00 97.8% 97.8% NA NA 

Underline: Bad HV Controller (ND); Bad Field Point (FD) 



ND Status 
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FD Status 
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Spectrum! 
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Events/POT + Intensity 
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